**Author's response to reviews**

**Title:** The combination therapy with alfacalcidol and risedronate improves the mechanical property in lumbar spine by affecting the material properties in an ovariectomized rat model of osteoporosis.

**Authors:**

Ayako Shiraishi (shiraishiayk@chugai-pharm.co.jp)
Sayaka Miyabe (miyabe@mat.eng.osaka-u.ac.jp)
Takayoshi Nakano (nakano@mat.eng.osaka-u.ac.jp)
Yukichi Umakoshi (umakoshi@mat.eng.osaka-u.ac.jp)
Masako Ito (masako@nagasaki-u.ac.jp)
Masahiko Mihara (miharamsh@chugai-pharm.co.jp)

**Version:** 5  **Date:** 13 May 2009

**Author's response to reviews:**

Dear Dr. Joseph Dunckley,

Thank you for e-mail. We understand your inquiries.

We revised the competing interests in our manuscript and uploaded the new version. Please confirm this matter.

Thank you very much for your kind consideration.

Sincerely yours,

* * * * * * * * * * * * * * * * * * * * * * * * * * * * *

Ayako Shiraishi, Ph.D.
Product Research Dept.
Chugai Pharmaceutical Co., Ltd.
1-135 Komakado, Gotemba, Shizuoka
412-8513 Japan
TEL : +81-550-87-6379
FAX : +81-550-87-5397
E-mail : shiraishiayk@chugai-pharm.co.jp
* * * * * * * * * * * * * * * * * * * * * * * * * * * * *